Author's response to reviews

Title: Cross-Sectional Study of Self-Reported Physical Activity, Eating Habits and Use of Complementary Medicine in Breast Cancer Survivors

Authors:

Arnoud Templeton (arnoud.templeton@uhn.ca)
Beat Thürlimann (beat.thuerlimann@kssg.ch)
Michael Baumann (michael.baumann@kssg.ch)
Michael Mark (michael.mark@ksgr.ch)
Sarah Stoll (sarah.stoll@kssg.ch)
Madeleine Schwizer (madeleine.schwizer@kssg.ch)
Daniel Dietrich (daniel.dietrich@sakk.ch)
Thomas Ruhstaller (thomas.ruhsstaller@kssg.ch)

Version: 3 Date: 5 February 2013

Author’s response to reviews: see over
To Christna Chap, PhD
Senior Executive Editor
BMC-series
BioMed Central
236 Gray's Inn Road
London WC1X 8HB

St. Gallen, February 4th 2013

Re: Revised MS 1988144254748640 “Cross-Sectional Study of Self-Reported Physical Activity, Eating Habits and Use of Complementary Medicine in Breast Cancer Survivors”

Dear Dr. Chap

We would like to thank the editors for their decision to accept our manuscript for publication.

As no further revisions were required by the reviewers, we have only added the name of the ethics committee who gave approval to the study and made some minor language and spelling corrections.

Again, we would like to thank you for publishing this manuscript in BMC Cancer.

Sincerely,

A. Templeton

Dr. Arnoud Templeton
on behalf of all authors